ഹ



JGR Space Physics

RESEARCH ARTICLE

10.1029/2021JA029801

Key Points:

- Probability distributions of Birkeland current densities are best-described by *q*-exponential distributions
- The probability of currents at any threshold is higher in the Northern Hemisphere
- Extreme currents are most likely on the dayside at a colatitude of 18°-22° (colocated with R2)

Correspondence to:

J. C. Coxon, work@johncoxon.co.uk

Citation:

Coxon, J. C., Chisham, G., Freeman, M. P., Anderson, B. J., & Fear, R. C. (2022). Distributions of Birkeland current density observed by AMPERE are heavy-tailed or long-tailed. *Journal* of Geophysical Research: Space Physics, 127, e2021JA029801. https://doi. org/10.1029/2021JA029801

Received 15 JUL 2021 Accepted 10 JAN 2022

Author Contributions:

Conceptualization: John C. Coxon, Gareth Chisham, Mervyn P. Freeman Data curation: Brian J. Anderson Formal analysis: John C. Coxon Funding acquisition: Robert C. Fear Investigation: John C. Coxon, Gareth Chisham, Mervyn P. Freeman Resources: Robert C. Fear Software: John C. Coxon, Gareth Chisham

Supervision: Robert C. Fear Visualization: John C. Coxon Writing – original draft: John C. Coxon Writing – review & editing: John C. Coxon, Gareth Chisham, Mervyn P. Freeman

© 2022. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Distributions of Birkeland Current Density Observed by AMPERE are Heavy-Tailed or Long-Tailed

John C. Coxon^{1,2} ^(D), Gareth Chisham³ ^(D), Mervyn P. Freeman³ ^(D), Brian J. Anderson⁴ ^(D), and Robert C. Fear² ^(D)

¹Department of Mathematics, Physics and Electrical Engineering, Northumbria University, Newcastle upon Tyne, UK, ²School of Physics and Astronomy, University of Southampton, Southampton, UK, ³British Antarctic Survey, Cambridge, UK, ⁴Johns Hopkins University Applied Physics Laboratory, Laurel, MD, USA

Abstract We analyze probability distributions of Birkeland current densities measured by the Active Magnetosphere and Planetary Electrodynamics Response Experiment. We find that the distributions are leptokurtic rather than normal and they are sometimes heavy-tailed. We fit *q*-exponential functions to the distributions and use these to estimate where the largest currents are likely to occur. The shape and scale parameters of the fitted *q*-exponential distribution vary with location: The scale parameter maximizes for current densities with the same polarity and in the same location as the average Region 1 current, whereas the shape parameter maximizes for current densities $|J| \ge 0.2 \ \mu A \ m^{-2}$ are most likely to occur in the average Region 1 current region, and second most likely to occur in the average Region 2 current region. However, for extreme currents ($|J| \ge 4.0 \ \mu A \ m^{-2}$), we find that the most likely location is colocated with the average Region 2 current region on the dayside, at a colatitude of $18^{\circ}-22^{\circ}$.

Plain Language Summary We use data from a telecommunications satellite network and measure currents flowing along the magnetic field lines above Earth's surface. We look at how strongly these currents flow above different parts of Earth's surface and plot a histogram of the strengths for each of those different locations. We then fit a mathematical model to the histograms to see how those histograms vary with location, and we are able to use the model to work out the probability of a current above a certain strength flowing. This is useful both to gain insights into the physics of the system, but also to gain insights into the potential impacts of these currents on infrastructure on Earth's surface.

1. Introduction

The way in which field-aligned currents, or Birkeland currents, are distributed in Earth's magnetosphere is critical to understanding the times at which Earth feels the largest effects from space weather. This is because Birkeland currents close through Hall and Pedersen currents flowing in Earth's ionosphere (Cowley, 2000). Ampère's law means that these currents cause magnetic perturbations that can be measured on Earth's surface, and Faraday's law in turn means that those magnetic perturbations induce currents on the surface, commonly known as geomagnetically induced currents (GICs). GICs are a key area of scientific interest and have been reviewed on multiple occasions (Ngwira & Pulkkinen, 2019; Pirjola, 2002; Thomson et al., 2009); they are also a key area of policy interest, due to the adverse impact of GICs on electrical power networks and the associated risk of high economic costs (Eastwood et al., 2018).

Several authors have previously examined the average behavior of Birkeland currents. Christiansen et al. (2002) reported maximum average current densities of ~0.2 μ A m⁻² observed by Ørsted and MAGSAT during quiet times as opposed to average densities of ~0.5 μ A m⁻² during disturbed times. Juusola et al. (2014) reported maximum average current densities of ~0.7 μ A m⁻² using CHAMP data during 2000–2010. Workayehu et al. (2019) used Swarm data from 2014 to 2017 and reported maximum means of 0.4 μ A m⁻² which is lower than Juusola et al. (2014) but consistent with Christiansen et al. (2002); this may indicate that CHAMP overestimated the current densities or may indicate that 2000–2010 was generally more active than 2014–2017, consistent with Solar Cycle 24 being generally quieter than the preceding solar cycle. Xiong et al. (2020) used DMSP data during 2010–2014, and reported the maximum average current density as ~0.2 μ A m⁻², which is lower than both Juusola et al. (2014) and Workayehu et al. (2019) and may indicate that 2010–2014 had lower activity levels than

2000–2010 and 2014–2017. Coxon et al. (2016) found that Northern Hemisphere currents were higher than those in the Southern Hemisphere during the Northern summer as well as spring and autumn; later studies found the same effect using Swarm (Workayehu et al., 2020) and DMSP (Xiong et al., 2020).

It was outlined in Cowley (2000) that the region 1 (R1) Birkeland current should be colocated with the open/ closed field line boundary (OCB). This has been confirmed experimentally, showing that Birkeland current morphology is intimately related to the location of the polar cap (Clausen, Baker, et al., 2013; Clausen et al., 2012; Clausen, Milan, et al., 2013; Coxon et al., 2014b). Studies have found that accounting for the resulting latitudinal variability can be very important (Coxon et al., 2017; S. E. Milan et al., 2015) and recently, a method has been made available to the community in order to allow current densities to be plotted in a coordinate system defined by the OCB (Burrell et al., 2020; S. Milan, 2019); this coordinate system shifts the OCB to 20° colatitude such that all current densities poleward of that are R1 current densities and equatorward of that are R2 current densities.

The probabilities of extreme GICs have been considered by measuring the rate of change *R* of the magnetic field that induces GICs at the Earth's surface and hence estimating where the largest GICs will occur (Rogers et al., 2020; Thomson et al., 2011) and what causes them (Freeman et al., 2019; Smith et al., 2019). Substorms are phenomena that cause large currents to flow into and out of the ionosphere in the substorm current wedge (e.g., Coxon et al., 2014b; 2017). However, Freeman et al. (2019) showed that only ~14% of the largest values of *R* correspond to magnetic perturbations observed during the substorm current wedge (termed DP-1), indicating that most of the highest *R* values are not directly attributable to substorms. Smith et al. (2019) performed a similar study with sudden commencements (SCs) of geomagnetic storms, finding that over 90% of the *R* values 3σ above the mean were observed within 72 hr of a SC, thus indicating that geomagnetic storm arrival predictions are key to predicting the impacts of space weather on Earth's surface. One method of examining the behavior of GICs is to model how they are distributed using a *q*-exponential function (Barbosa et al., 2017), which has also been investigated in some detail in ionospheric vorticity using SuperDARN data (Chisham et al., 2009; Chisham & Freeman, 2010, 2021).

Chisham et al. (2009) showed that ionospheric vorticity measured with the SuperDARN radars (Chisham et al., 2007) is closely related to Birkeland current behavior, as expected theoretically (Freeman et al., 1990; Sofko et al., 1995). Chisham et al. (2009) utilized SuperDARN measurements from 2000 to 2005 inclusive in order to obtain enough vorticities to evaluate the distributions of the ionospheric vorticities. Chisham and Freeman (2010) showed that radar-observed ionospheric vorticities demonstrated heavy-tailed distributions, finding that the best fit to the distributions was a Weibull or a q-exponential (Tsallis) distribution rather than a Gaussian. These distributions have higher kurtosis than a Gaussian distribution (meaning they are leptokurtic) and often have higher kurtosis than an exponential (meaning they are heavy-tailed); this implies that extreme ionospheric vorticities are more common than would be expected if the data were distributed normally or exponentially. Although Chisham et al. (2009) demonstrated that ionospheric vorticity could be used as a proxy for Birkeland current, the work done to quantify the vorticity distributions found using SuperDARN data (Chisham & Freeman, 2010, 2021) has not been done using Birkeland current data directly, because until now a suitable data set was not available for this type of analysis. We address this by using the Active Magnetosphere and Planetary Electrodynamics Response Experiment (AMPERE, Anderson et al., 2000, 2021; Waters et al., 2001, 2020). Furthermore, SuperDARN studies of ionospheric vorticity have been constrained by the geographical locations of the radars, whereas AMPERE is truly global.

Recently, Chisham and Freeman (2021) showed how the parameters of a *q*-exponential fit to the ionospheric vorticity distribution vary spatially across the high-latitude ionosphere, and used these parameters to calculate the probabilities of seeing extreme vorticities in different parts of the polar cap and auroral region. The *q*-exponential is based on two parameters: *q* and κ . (The *q*-exponential is described in detail in Section 3, below.) *q* here is a free parameter which is known to be associated with intermittency; extreme values of a variable are much more likely than would be expected from a Gaussian distribution if that variable has an intermittent distribution. Higher values of *q* indicate more intermittency. Intermittency is one property of turbulence, and examinations of ionospheric vorticity in the paradigm of turbulence have shown that intermittency is similar between the ionosphere and solar wind (Abel et al., 2007) and that this only holds true when the two regimes are coupled by magnetic reconnection during Southward IMF (Abel et al., 2009). Chisham and Freeman (2010, 2021) argued that the fact that ionospheric vorticity was well-modeled by non-Gaussian distributions such as the *q*-exponential provided evidence that there is a spatial variation in intermittency and therefore a spatial variation in the likelihood of



Figure 1. Maps showing the mean current densities \overline{j} (left) and mode current densities j_m (right) in the Northern Hemisphere (top) and the Southern Hemisphere (bottom) for 2010–2012. Circles denote co-latitude and perimeter numbers denote magnetic local time in altitude-adjusted corrected geomagnetic coordinates.

extreme ionospheric vorticities. Our motivation for investigating the distributions of Birkeland currents is to confirm whether they have this property, and if so, how it is spatially distributed.

In this paper, we use AMPERE data from 2010 to 2012 (Section 2) to evaluate the distributions of the Birkeland currents (Section 3), before using those distributions to evaluate the probabilities of seeing Birkeland current densities of varying magnitudes at mid-high latitudes (Section 4). We discuss the results (Section 5) before concluding (Section 6). The reader should note that while Figure 1 plots maps of current density, the remainder of the maps plotted in the subsequent sections are not plots of current density.

2. Birkeland Current Densities From AMPERE

AMPERE is a data set that combines measurements of magnetic perturbations from Iridium satellites with spherical harmonic fitting to estimate the spatial variation of Birkeland current densities. The data processing method was first outlined by Anderson et al. (2000); Waters et al. (2001) and a more detailed consideration is presented in Anderson et al. (2021); Waters et al. (2020). This current density *j* is presented on a grid of 1° latitude by 1 hr of magnetic local time (MLT) in altitude-adjusted corrected geomagnetic coordinates (AACGM), with a convention that upward currents are *j* > 0 and downward currents are *j* < 0. *j* is estimated from measurements in a ten-minute long sliding window, which is evaluated every 2 min. AMPERE has been used widely to study Birkeland current



Maps of standard error (2010-01-01 to 2012-12-31)



Figure 2. Maps showing the standard error on the mean for the Northern Hemisphere (top) and the Southern Hemisphere (bottom) for 2010–2012.

densities (Coxon et al., 2018). In this study, we use AMPERE data from 2010 to 2012; this is because there are some temporal gaps in AMPERE data in subsequent years which introduce uneven sampling of seasons.

Figure 1 (left) shows the spatial variation of the resulting mean current density \overline{j} at each latitude-MLT measurement location, which looks very similar to previous images of the mean Birkeland current densities. The morphology of these currents is as follows (Iijima & Potemra, 1976a, 1976b, 1978): The means in both hemispheres comprise two rings of current, one poleward (Region 1, or R1) and one equatorward (Region 2, or R2). The R1 currents are downward on the dawn side and upward on the dusk side, and the R2 currents are oppositely directed on the dawn and dusk sides. The regions are statistically latitudinally broad, spanning up to 10° of latitude from their poleward to equatorward extent. The mean j in the Northern Hemisphere is stronger than in the Southern Hemisphere (Coxon et al., 2016; Workayehu et al., 2020; Xiong et al., 2020). The center of the two rings of current is displaced toward the nightside. There are also two regions of mean current within the polar cap, which are some combination of Region 0 and NBz currents. The mean currents are in excess of 0.1 µA m⁻², consistent with the quiet time averages reported by Christiansen et al. (2002); Xiong et al. (2020) but lower than the averages reported by Juusola et al. (2014); Workayehu et al. (2019); this may be because 2010-2012 was in general less active than the periods reported by those authors, or may be an indication that AMPERE is underestimating the current densities.

Figure 1 (right) shows the mode current density j_m . We again see rings of current which are positive/negative on different sides of the polar cap, but these are poleward of the \bar{j} of the same polarity. j_m is smaller than \bar{j} , and the regions of current are latitudinally narrower, extending ~5° in latitude. As with \bar{j} , the rings are displaced toward the nightside. In cases where j_m and \bar{j} are of different signs, this tells us about the current distributions—for example, at 6 hr of MLT and 18° colatitude, where $j_m > 0$ but $\bar{j} < 0$, this implies that the most common current density is positive, but the negative current densities are typically larger.

Previous work has shown that the currents reported by AMPERE in the polar cap are subject to large uncertainties (D. J. Knipp et al., 2014). To examine this, we present Figure 2 which shows the standard error on the mean for the same spatial regime as Figure 1. The standard error is higher in the Northern Hemisphere than in the Southern Hemisphere, which we infer is a signature of the generally higher Northern current densities. The standard error is higherst colocated with the boundary between the R1 and R2 current regions and the NBz region shown in Figure 1, which is consistent with the work of D. J. Knipp et al. (2014). However, since we have approximately 7×10^5 data per

bin, the standard errors are of the order 10^{-7} , which is substantially lower than the means and modes reported in Figure 1; as such, we conclude that the error on the data presented herein is small.

The *q*-exponential is positive definite with a maximum at the origin. This means that it can be fit to any data distribution in which all the data are of the same sign (Chisham & Freeman, 2010). As AMPERE data can be positive or negative in any given bin, we take the mode in each bin (plotted in Figure 1, right) and subtract it from the distribution of current densities in each bin prior to performing the fitting. This shifts the distribution such that the mode is at zero, and means that we can fit two *q*-exponentials with a maximum at the origin (zero) and current densities of the same sign on each side of the origin. In this way we fit the *q*-exponential separately to the distribution of the modulus of the measured "positive" current density ($j > j_m$) and to the modulus of the "negative" current density ($j < j_m$). We refer to $j > j_m$ as positive current density and $j < j_m$ as negative current density hereafter to aid readability. We plot the number of AMPERE estimates of *j* in each bin across the polar cap for the







Figure 3. Maps showing the number of positive current densities $(j > j_m)$ left column) and negative current densities $(j < j_m)$ right column) in the Northern Hemisphere (top row) and the Southern Hemisphere (bottom row) for 2010–2012.

Northern (top) and Southern (bottom) Hemispheres and for positive current densities (left) and negative current densities (right) in Figure 3.

The regions in which the number of data is lower or higher than the average correspond well to the \bar{j} plotted in Figure 1 (left). We note that this does not show the number of satellites contributing to the fits, but rather the number of fits in each bin, such that the number of fits is equal at all locations for total positive plus negative measurements. We also note that there are over 0.2 million data per bin, which is typically much larger than the number of data when using SuperDARN (Chisham & Freeman, 2021).

3. Distributions of Birkeland Current Densities

In this study, we fit q-exponential distributions to the Birkeland current densities from AMPERE, following the logic of previous work on the underlying ionospheric vorticities (Chisham et al., 2009; Chisham & Freeman, 2010, 2021). The q-exponential probability density function (PDF) is given by

$$f(j) = \frac{1}{\kappa} \left(1 - \frac{(1-q)j}{\kappa} \right)^{q/(1-q)}$$
(1)

where $f_{q,\kappa}(j)$ is the probability density, κ is the scale parameter, q is the shape parameter, and j was defined earlier as current density. Values of q < 1 have less kurtosis than an exponential fit, indicating that the distribution decays





Figure 4. Plots showing how variations in q (left) and κ (middle and right) affect the shape of the distribution. The color bars refer to the value of q and κ . The value at which the other variable is held is shown in the top-right. The middle and right-hand panels show the same trends, but the right-hand panel shows lines overlaid; horizontal lines are plotted at $f(j) = f(j = 0) \times q^{-35}$ and vertical lines are plotted at the corresponding j value. This demonstrates that the shape of the distribution is dictated by q and the scale of the distribution is dictated by κ .

more quickly than an exponential would. Values of q > 1 have more kurtosis, and so decay more slowly; they are heavy-tailed. The κ parameter governs the scale of the distribution; the decay rate is constant with respect to variations in κ but higher κ implies that the vorticity varies more from the mode magnitude of the current density and low κ means the variability is lower. Therefore, both higher q and higher κ imply that more extreme current densities will occur, but the underlying reason for this are higher intermittency and higher variability respectively.

Figure 4 shows the way in which the shape of a *q*-exponential distribution changes with the two key parameters *q* (left) and κ (middle, right). In these schematics the value of the other parameter is kept constant ($\kappa = 0.1$ and q = 1.1 respectively). Where q = 1, the *q*-exponential tends to the exponential distribution; where q < 1 the distribution is less heavy-tailed than the exponential (it decays more quickly), whereas at larger *q* the distribution becomes heavier-tailed (it decays more slowly). Increases in κ make the distribution longer-tailed or wider, which does not affect the decay and therefore does not affect the shape of the distribution. This is shown by Figure 4 (right) which shows the value of *j* at which the value of *f*(*j*) has decreased to a value of $f(j) = f(j = 0) \times q^{-35}$ (chosen to be visible on the schematic); it can be seen that as kappa increases monotonically so does *j*, showing that the distribution is being scaled upward (and thus becoming longer-tailed) with increasing κ .

Figure 5 shows how current densities are distributed in two of the coordinates underlying Figure 1. We use maximum likelihood estimation (MLE) to determine the best fit PDF to the distributions of current density (Chisham & Freeman, 2010, 2021). Figure 5 suggests that the *q*-exponential fit is better than the exponential or Gaussian fits to the distribution for both positive and negative current densities; this is confirmed by comparing the Akaike Information Criteria (AIC) of the three fits.

Figure 5 (top) shows the distribution for a colatitude of 15° and an MLT of 18 hr. Both the mode and mean are positive, and the mean is larger than the mode. This coordinate is chosen because it is in the region of average R1 current on the dusk side of the polar cap, and demonstrates the way in which the asymmetry and the tails of the underlying distributions dictate whether $\bar{j} > j_m$ or $\bar{j} < j_m$ in a given cell. The tails of the R1 current distribution decay more slowly than a Gaussian (they are leptokurtic) but with approximately the same decay as an exponential (they are not heavy-tailed). The top of the distribution is wider than an exponential, especially for positive current densities. These inferences are consistent with the values of the fit, which shows that κ is high (implying a wide or long-tailed distribution) and that q < 1 (implying that the distribution decays more quickly than an exponential).

Figure 5 (bottom) shows the distribution for a colatitude of 155° and an MLT of 6 hr. This coordinate is chosen to demonstrate average R2 current on the dawn side of the polar cap. This coordinate displays positive current

10.1029/2021JA029801





Figure 5. The distribution of Birkeland current densities for 2010–2012 inclusive as observed by AMPERE (black) for (top) 15° colatitude and 18 hr magnetic local time (MLT), and (bottom) 155° colatitude and 6 hr MLT. Overlaid lines show the fits to the distributions: exponential (orange), *q*-exponential (blue) and Gaussian (green). The *q*-exponential fits better to the distribution for negative and positive currents, and the parameters of the fit are presented in the top-left and top-right respectively. The solid gray line is the mode and the dotted black line is the mean.

density on average ($\bar{j} > j_m$), but we see that there are some large negative current densities within the distribution. There are fewer negative current densities than positive for any given magnitude, and the logarithmic scale makes it obvious that the vast majority of the negative current densities in this coordinate are in the range $-0.4 < j < 0 \ \mu A \ m^{-2}$: that is to say the R2 distribution is narrower than the distribution for R1, but clearly lepto-kurtic and heavy-tailed compared to the exponential. These inferences are also consistent with the *q*-exponential fit, which shows that q > 1 (implying that the distribution decays more slowly than an exponential) but shows that κ is small (implying a narrow distribution).

Figures 6 and 7 show the values of q and κ respectively for the distributions in each bin for the period of interest. Light gray bins correspond to where q = 1. Because neither q nor κ are negative, we multiply them by the sign of the mean of the currents \overline{j} when plotting them to make it easier to compare them to the large-scale current systems shown in Figure 1 (left).

Figure 6 shows the values of q. The locations of q > 1.2 are mirrored for positive and negative current densities, which are the mirror image of one another across the noon-midnight meridian, but which are in almost identical locations in the Northern and Southern Hemispheres; as such, we focus on the Northern Hemisphere henceforth. Since high values of q are colocated with negative \overline{j} on the dusk side and positive \overline{j} on the dawn side, we infer that high values of q are predominantly associated with the average R2 current system. For positive current densities (left), the region of high q on the dawn side is further from the pole than the region on the dusk side; the dusk side region encroaches onto the equatorward edge of the R1 current system. The distribution at a given location will be a mixture of the distributions of the processes which occur in that location (Chisham & Freeman, 2021), and so we interpret this as an effect of the mixing of the R1 and R2 distributions due to the expansion and contraction of





Figure 6. Maps showing the value of q in the best q-exponential fit to the underlying distribution per coordinate, in the same format as Figure 3. White bins indicate bins in which the fitting routine to identify the best parameters for the q-exponential was unsuccessful (note that the color bar is gray at 0, rather than white). The values are multiplied by the sign of the mean current \overline{j} , such that positive values correspond to positive \overline{j} and vice versa. Note that these maps are not maps of current density.

the current ovals in response to flux transfer into and out of the polar cap (Cowley & Lockwood, 1992; Lockwood et al., 1990; Lockwood & Freeman, 1989; Siscoe & Huang, 1985). We see the mirror image of this for negative current densities (right). This implies that the R2 currents are characterized by high intermittency and therefore the distributions are heavy-tailed, with a decay rate lower than an exponential distribution. The R1 currents do not display high intermittency and instead the distributions are close to exponential.

At higher latitudes in the Northern Hemisphere, for positive current densities (left) there are regions of $q \sim 1.1$ on the dayside poleward of 10°, with negative \bar{j} on the dusk side and positive \bar{j} on the dawn side. These appear to be associated with NBz currents which we associate with northward IMF. For negative current densities (right) there is a region of $q \sim 1.2$ poleward of 10° pre-noon and equatorward of 10° post-noon. The same is seen with slightly lower q in the Southern Hemisphere. We associate these four high q regions with the NBz current system caused by northward IMF B_Z ; these current systems do not have intermittency as high as the R2 current systems but are heavy-tailed.

Figure 7 shows the values of κ . The locations of $\kappa > 0.1$ are mirrored for positive and negative current densities and are very similar between the two hemispheres; as such, we focus again on the Northern Hemisphere. For positive current densities (left), κ is highest ($\kappa > 0.2$) for the positive \overline{j} on the dusk side, corresponding to the R1 current system, but encroaches very slightly into the R2 current system. For negative current densities (right) we see similar morphology of the opposite polarity. There is also a region of $\kappa \sim 0.1$ corresponding to the poleward





Figure 7. Maps showing the value of κ in the best *q*-exponential fit to the underlying distribution per coordinate, in the same format as Figure 6. The values are multiplied by the sign of the mean current \overline{j} , such that positive values correspond to positive \overline{j} and vice versa. Note that these maps are not maps of current density.

edge of positive currents on the dawn side, corresponding to the edge of the R2 current system. Again, we interpret this encroachment into R2 as evidence of a mixture distribution due to the motion of the current ovals. The high values of κ imply that although the R1 current distribution is not heavy-tailed, it is wide or long-tailed. $\kappa > 0.2$ on either side of noon at high latitudes (<10°) corresponding to the NBz current system, from which we infer that NBz currents have a long-tailed or wide distribution. Signatures associated with NBz are more difficult to discern in the Southern Hemisphere (bottom).

As such, we have concluded that R1 current is characterized by long-tailed or wide (high- κ) but not heavy-tailed ($q \sim 1$) distributions and R2 current is characterized by narrower distributions (low κ) which are heavier-tailed (high q). NBz currents sit in the middle in both respects; heavier-tailed than R1 but less than R2, and longer-tailed than R2 but less than R1. This implies that current densities closer to the center of the distribution are more likely in R1 than in R2, but that very extreme currents are more likely to be seen in R2 current. It further implies that moderate current densities are less likely in NBz but extreme current densities are also less likely, which is consistent with Figure 1 showing that the average NBz currents are smaller than R1 and R2. We examine this in more detail in the next section.



4. Probabilities of Birkeland Current Densities

Having calculated f(j) per coordinate in Section 3, we can now calculate the probability of extreme values of current density by using the survival function (Chisham & Freeman, 2021). For $j > j_m$ the survival function is given by

$$P_{+}(J \ge j) = \frac{n_{+}}{n_{+} + n_{-}} \left(1 - \frac{(1 - q_{+})(j - j_{m})}{\kappa_{+}} \right)^{1/(1 - q_{+})}$$
(2)

and for negative currents $(j < j_m)$, the survival function is given by

$$P_{-}(J \le j) = \frac{n_{-}}{n_{+} + n_{-}} \left(1 - \frac{(1 - q_{-})(j_{m} - j)}{\kappa_{-}} \right)^{1/(1 - q_{-})}$$
(3)

where *P* is the probability of current flowing and *n* is the number of measurements. P_+ , n_+ , q_+ , κ_+ are those quantities for positive current densities and P_- , n_- , q_- , κ_- are those quantities for negative current densities.

We assess the probabilities of seeing current densities above 0.2, 1.0, 2.0, and 4.0 μ A m⁻² in the following figures, and discuss the probabilities we find and our interpretations of these probabilities in the context of the scientific literature in Section 5.

Figure 8 shows the probability $P(|J| \ge 0.2 \ \mu A \ m^{-2})$, which is sometimes used as the bottom of the color scale when plotting current densities with AMPERE to highlight large-scale structure (e.g., Coxon et al., 2014a). Qualitatively, the probabilities presented here show that the highest probabilities are co-located with the NBz, R1, and R2 current regions seen in Figure 1 (left) and are also co-located with the regions of high q or κ described in Section 3, which is as we would expect. Quantitatively, $P(|J| \ge 0.2 \ \mu A \ m^{-2})$ is almost 30% for Region 1 current in both hemispheres, and is closer to 15% for Region 2 and NBz current. Thinking back to the discussion of q and κ in Section 3, the fact that these current densities are more probable in the R1 current region is consistent with the fact that R1 is characterized by wider or longer-tailed (higher- κ) distributions than R2 current; moderate current densities are more probable in R1 as a result.

Notably, the chance of seeing current $P(|J| \ge 0.2 \ \mu \text{A m}^{-2})$ is higher in the Northern Hemisphere than in the Southern Hemisphere, especially toward the dayside. In the Northern Hemisphere the probabilities for R1 current decrease on the nightside compared to the dayside; this does not occur in the Southern Hemisphere, for which the probabilities for R1 current stay close to 30% at all dusk local times for positive *j* and all dawn local times for negative *j*.

Figure 9 shows $P(|J| \ge 1.0 \ \mu \text{A m}^{-2})$. The difference between this and Figure 8 is striking. We see two main zones of likelihood P > 0; for positive current densities these zones are located between noon and 22 MLT (on the dusk side) and between midnight and 6 MLT (on the dawn side). We refer to these zones as Zone A and Zone B respectively hereafter, and these zones are highlighted in Figure 12 for the convenience of the reader. For negative current densities Zones A and B are mirrored, such that Zone A is on the dawn side (2 MLT to noon) and Zone B is on the dusk side (15–21 MLT). The probability $P(|J| \ge 1.0 \ \mu \text{A m}^{-2})$ in Zone A is $\sim 1\%$ ($\sim 30\times$ lower than in the previous figure) and in Zone B is $\sim 0.1\%$. The hemispheric difference here remains the same as in Figure 8, in that Northern Hemisphere likelihoods are higher than those in the Southern Hemisphere.

Figure 10 shows $P(|J| \ge 2.0 \ \mu \text{A m}^{-2})$, which looks qualitatively similar to Figure 9. The key difference is that the probability in Zone A is now ~0.1% and in Zone B is $P \sim 0.01\%$, which is an order of magnitude smaller than in the previous figure. The Northern Hemisphere probabilities are 2× higher than the Southern Hemisphere. The probabilities in both regions again decrease by an order of magnitude in Figure 11, which shows $P(|J| \ge 4.0 \ \mu \text{A m}^{-2})$ and which looks very similar to Figure 10.

5. Discussion

The mean currents presented in Figure 1 (left) imply that R1 currents are typically larger than R2 currents, consistent with previous averages (Anderson et al., 2008; Weimer, 2001) but potentially attributable to spatial smearing at more equatorward colatitudes (Coxon et al., 2017). Studies have found that R1 currents are typically stronger than R2 when the amount of current flowing is high, but have found that R2 currents dominate when the





Figure 8. Maps of $P(|J| \ge 0.2 \ \mu \text{A m}^{-2})$, in the same format as Figure 6. Bins in which the probability could not be computed were set to zero. Note that these maps are maps of probability and not maps of current density.

amount of current is lower (Coxon et al., 2014a); Figure 8 shows that the probability of seeing current densities $P(|J| \ge 0.2 \ \mu \text{A m}^{-2})$ is higher for R1 (10°–15° colatitude) than it is for R2 (18°–25°), which is consistent with this.

5.1. Correspondence to Ionospheric Vorticity

Chisham et al. (2009) reported that the large-scale morphology of ionospheric vorticity is similar to the expected large-scale morphology of Regions 1 and 2 Birkeland current. Plotting the mean ionospheric vorticity on a grid identical to the ones in this manuscript (but with an equatorward cutoff at 24° colatitude) showed two large regions between $10^{\circ}-20^{\circ}$ colatitude on either side of the noon-midnight line corresponding to R1 current and two regions of the opposite sign corresponding to R2 current between $20^{\circ}-24^{\circ}$ colatitude (the R2 vorticity regions extend to the equatorward edge of the grid and so may extend further equatorward than that range implies). Additionally, Chisham et al. (2009) observed regions of opposite polarity to the poleward edge of the R1 regions; they subdivided by IMF clock angle to differentiate these into NBz and R0 current systems.

Figure 1 shows that the mean Birkeland current measured by AMPERE is broadly similar to that reported by Chisham et al. (2009), with the R1 and R2 currents approximately colocated with their corresponding vorticity regions. The one area in which this was not the case was on the dayside, where our R1 and R2 currents are slightly closer to the pole than the vorticity regions; however, this could be due to the different time domains of





Figure 9. Maps of $P(|J| \ge 1.0 \ \mu \text{A m}^{-2})$, in the same format as Figure 8. Note that these maps are maps of probability and not maps of current density.

the two datasets. We see mean current densities in the polar cap consistent with the NBz vorticities observed by SuperDARN.

Chisham and Freeman (2021) presented the values of the mode for the SuperDARN vorticities and reproduced the means shown in Chisham et al. (2009). The mode values reported were approximately half the mean values, which is consistent with the values reported from AMPERE in Figure 1. However, the morphology of the mode vorticities is quite different from AMPERE, with the mode values associated with R1 current smaller than those associated with R2 current, as opposed to the values reported by AMPERE in which the modes in the R1 current are larger than those in R2. In the AMPERE data we also see mode currents which are of the same polarity as the mean NBz currents, whereas there is no signature of the NBz vortices in the SuperDARN modes reported by Chisham and Freeman (2021).

In terms of data coverage, Chisham et al. (2009) reported that there were between 100 and 100,000 data per bin in the colatitude range $4^{\circ}-24^{\circ}$ and zero data per bin outside that range (Chisham et al., 2009). The data coverage was uneven due to the uneven nature of SuperDARN observations: there is a region in which most of the data is found which is a circle offset from the pole toward the nightside, such that most of the data is found in the colatitude range $17^{\circ}-23^{\circ}$ at midnight; $15^{\circ}-20^{\circ}$ at dawn and dusk; and at $10^{\circ}-15^{\circ}$ at noon. The region with the most data is this region on the nightside; the region with the least is the equatorward edge on the dayside and the polar edge at all magnetic local times. Contrasting this to Figure 3 herein, we can see that there are ~700, 000 data per bin (the sum of the two panels), and the number of positive or negative current densities per bin varies between ~200, 000 and ~550, 000, which is between twice the number and 55,000 times the number of data per





Figure 10. Maps of $P(|J| \ge 2.0 \ \mu \text{A m}^{-2})$, in the same format as Figure 8. Note that these maps are maps of probability and not maps of current density.

bin. Additionally, the figures in this paper extend to 40° instead of cutting off at 24° , and in this paper we are able to examine the Birkeland currents in the Southern Hemisphere whereas this has not yet been possible with SuperDARN (Chisham et al., 2009; Chisham & Freeman, 2010, 2021).

The vorticities observed using SuperDARN data have been shown to be generally well-described by fitting a q-exponential distribution to each side of the distribution from the mode (Chisham & Freeman, 2010). The best values for the distribution in each bin were examined by Chisham and Freeman (2021), corresponding to the top rows of our Figures 6 and 7. Interestingly, their results differ somewhat from the ones presented in our figures. They found, like we have, that the Region 2 current system is characterized by high values of q but low values of κ , and found that dayside R1 currents are close to exponential ($q \sim 1$ and high κ), similar to our findings. However, they found that the characteristic distribution for R1 was different on the nightside, which showed higher values of both parameters; we do not observe this. They also found that the polar cap was characterized by intermediate values of q (which we observe herein) but by low values of κ (which we do not observe). There will be some differences between the analyses because the current is measured in space in the inertial frame, whereas the vorticity distributions must mainly be related to spatial changes in conductance, and this is an area for future research.

In terms of the latitude ranges, our finding is similar to Chisham and Freeman (2021), who found that large vorticities were most likely between colatitudes of $15^{\circ}-20^{\circ}$ but that extreme vorticities were most likely at colatitudes equatorward of 20° . They found that the largest probabilities were between 10 and 18 MLT, which is similar to





Figure 11. Maps of $P(|J| \ge 4.0 \ \mu \text{A m}^{-2})$, in the same format as Figure 8. Note that these maps are maps of probability and not maps of current density.

our result in that it is biased toward dusk, but they did not see a bimodal probability distribution in MLT sector as we do.

5.2. Spatial Probability Variations

Figures 8–11 show that the largest probabilities of observing a current above a given threshold occur at 05–09 MLT (for negative current density) and 14–20 MLT (for positive current density) and that the peak probability moves equatorward with increasing threshold. This movement is shown more clearly in Figures 13 and 14. Figure 13 shows cuts of the probability through the meridian of MLT = 15 and 3h for positive current densities and through MLT = 21 and 9 hr for negative current densities. Zone A corresponds to the 15 and 9 hr meridians, whereas Zone B corresponds to the 3 and 21 hr meridians. Figure 14 then plots how the colatitude of peak probability varies with current threshold for zones A and B. In both figures, we can see that the colatitude of peak probability moves equatorward as the threshold current density is increased.

Figure 12 also shows that the peak current is most likely to be seen in the Northern Hemisphere, as has been previously found for total current (Coxon et al., 2016) and Poynting flux (D. Knipp et al., 2021; Pakhotin et al., 2021). At the lowest thresholds (Figures 13a and 13b) the Northern Hemisphere currents are \sim 50% larger than those in the Southern Hemisphere, and Northern Hemisphere dominance increases with current threshold (\sim 7× larger at the highest thresholds, shown in Figures 13g and 13h).

14 of 20





Figure 12. Key to interpreting zones in Figures 9, 10 and 11.

5.3. Interpreting the Probabilities in Terms of R1 and R2 Current

Zone A occurs in the same sense as Region 1 current; that is to say, for positive current densities it is on the dusk side, and for negative current densities it is on the dawn side. R1 current is positive on the dusk side and negative on the dawn side. It is tempting to interpret this as evidence for this Zone comprising R1 current. However, as we have discussed previously in this study, the large-scale sense of R1 and R2 current does not preclude large current densities on the "wrong" side of the distribution. For example, in Figure 5, the R1 current (top) and R2 current (bottom) are taken from bins which are plotted as positive current density in Figure 1 because the mean of both distributions is positive. However, there is clearly a lot of negative current densities contributing to both distributions, and some of them are large (especially in Figure 5, bottom). Therefore, it is not implausible that there are (at times) very large negative current densities in a spatial region which is characterized on average by positive current.

As such, we can explain Zone A in one of two paradigms, both of which have interesting repercussions for the behavior of the system.

5.3.1. Paradigm 1

At any given location, we will measure either R1 or R2 and so the negative and positive sides of the distribution are driven by R1 and R2 dominating at a location at different times. It has been shown that the current ovals expand and contract with the polar cap (Clausen, Baker, et al., 2013; Clausen et al., 2012; Clausen, Milan, et al., 2013) as part of the expanding/contracting polar cap (ECPC) paradigm (Cowley & Lockwood, 1992;



Figure 13. A figure showing cuts through the 15 and 3 hr magnetic local time (MLT) meridian for positive current densities (left-hand column) and 21 and 9 hr MLT meridian for negative current densities (right-hand column). The rows indicate different thresholds of P(|J|): (a and b) $P(|J| \ge 0.2 \ \mu \text{A m}^{-2})$; (c and d) $P(|J| \ge 1.0 \ \mu \text{A m}^{-2})$; (e and f) $P(|J| \ge 2.0 \ \mu \text{A m}^{-2})$; (g and h) $P(|J| \ge 4.0 \ \mu \text{A m}^{-2})$. The dashed lines refer to the colatitude of peak probability for each panel. Blue indicates the Northern Hemisphere and orange the Southern Hemisphere.





Figure 14. Graph showing the colatitudes of peak probabilities from Figure 13. The left-hand column shows the Northern Hemisphere and the right-hand column shows the Southern Hemisphere. Panels (a and b) show the location of peak probability of the Zone A plotted in Figure 13 and panels (c and d) show the Zone B meridians. The values for the positive current densities are plotted in red and have 0.1° added; for negative current densities, blue is used and 0.1° is subtracted. The additions and subtractions are to make it easier to observe the trend for the reader.

Lockwood et al., 1990; Siscoe & Huang, 1985), and thus the current ovals are at their most expanded when the amount of magnetic flux (and therefore energy) in the system is at its highest.

It has further been shown that more energy is deposited by substorms when more magnetic flux is present in the polar cap (Coxon et al., 2014b; S. E. Milan et al., 2007) and that more current flows during and immediately after magnetic reconnection (Anderson et al., 2014; Coxon et al., 2014a, 2017). As such, if the first paradigm is the explanation behind our observations, we can interpret this in terms of more energy and more current flow during periods of magnetic reconnection. In this paradigm, the peak probability of extreme current results from R1 current which has moved equatorward as magnetic flux is opened by dayside reconnection, which is consistent with the sense of the currents.

However, this model implies that the intermittency of R1 current increases (i.e., q increases) depending on the size of the polar cap. We would not expect the underlying distribution of each current region to change in this way, and this result would be unintuitive.

5.3.2. Paradigm 2

In any given current system, the large-scale current system will comprise multiscale currents which have fine structure. There can be both intense upward and downward currents, but these average to look like the R1 and R2 current systems. In this model, R1 distributions are characteristically wide or long-tailed (high κ) and R2 distributions are characterized by high intermittency and therefore heavy-tailed (high q); the probabilities at high thresholds are related to distributions with high intermittency.

We find that the strongest currents are seen in the R2 system, and we find that the strongest currents are seen at dawn and dusk, biased toward the dayside. Coxon et al. (2014b) showed that the ratio of R1 to R2 current is larger during substorms than during other times, and Coxon et al. (2017) showed that although R2 contributes to the substorm current wedge it is to a smaller extent than R1. Additionally, if these signatures are due to the substorm current wedge they would occur on either side of midnight; therefore, we infer that these very strong currents are not seen during substorms.

It has long been thought that R2 currents close through the partial ring current on the nightside of Earth (Iijima & Potemra, 1976a, 1978), and it is well-known that the ring current is enhanced during geomagnetic storms (Chapman & Ferraro, 1933). As such, if paradigm 2 is underpinning our observations, we infer that highest probabilities





Figure 15. Maps of $P(|J| \ge 2.0 \ \mu A \ m^{-2})$, in the same format as Figure 8 but expressed in coordinates defined by the location of the open-closed field line boundary (Burrell et al., 2020). Note that these maps are maps of probability and not maps of current density.

occur when R2 is closing through the enhanced ring current during geomagnetic storms and that these closures happen around the dawn and dusk sectors.

However, in this paradigm it is unintuitive that the largest R2 currents occur in the opposite sense to the average R2 current system. We would expect the strongest currents to occur on the side of the distribution that dominates the mean current of a given region, although it may be that our observations are consistent with the embedded Birkeland currents recently reported by Liu et al. (2021). In a future study, we will repeat this analysis based on times at which we see ring current intensifications in order to verify this inference.

5.3.3. Why Not Both?

It is possible that both paradigms 1 and 2 are true, and that our observations are the result of an interplay between the two. We also need to note that AMPERE does not have fine-scale spatial resolution, and so it may be that both are true but AMPERE observations bias the observer toward interpreting the data in paradigm 1.

Figure 15 shows the data shifted into a coordinate system defined by the location of the open-closed field line boundary (OCB) (Burrell et al., 2020; S. Milan, 2019; S. E. Milan et al., 2015) in order to try to address this ambiguity. In this plot the OCB is located at 20°, such that R1 currents should be poleward and R2 currents should be equatorward. For positive current densities, the region of highest probability is equatorward of the circle, which may suggest that the largest currents observed here are R2 currents (paradigm 2). However, for negative current densities, the region of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle, which may suggest that the largest currents of the circle of the circle, which may suggest that the largest currents of the circle o



observed here are R1 currents (paradigm 1). This confirms the need for further work which examines the distributions of the currents during different types of driving.

6. Conclusion

We take maps of AMPERE data and analyze them statistically. We find that the mean current densities resemble the Regions 1 and 2 current systems which have been investigated throughout the history of the field (Birkeland, 1908; Iijima & Potemra, 1978; Liu et al., 2021; Weimer, 2001). However, we find that the modes of the current density distributions are displaced toward the pole relative to the means, and that the mean current densities traditionally interpreted as R1 and R2 are explained mainly by asymmetries in the distribution of the currents rather than by the mode current.

We analyze the current density distributions and find that they are well-described by a *q*-exponential distribution rather than a Gaussian distribution. We infer that the current densities exhibit complex behavior due to their heavy-tailed distributions, and we fit q-exponential distributions to describe the behavior of the distribution.

From the fits we derive the probability of current densities above certain thresholds and find that R1 currents are relatively more likely to be of clearly measurable magnitude than R2 currents, and that currents are most likely to occur at colatitudes of 10° -15°. However, the most extreme currents are seen at 18° -22° and are potentially linked to one of two paradigms:

- 1. The most extreme currents occur in the R1 currents at the point when the current ovals are most expanded.
- 2. The most extreme currents occur in the R2 currents due to closure through an intensified ring current during geomagnetic storms.

Adjusting the coordinate system to one defined by the OCB implies that these paradigms both contribute to the observations, and further work is needed to understand the contributions of each.

Data Availability Statement

The Iridium-derived AMPERE data used in this paper can be obtained from the AMPERE Science Center at http://ampere.jhuapl.edu/. The OMNI data used in this paper can be obtained from NASA/GSFC's Space Physics Data Facility's CDAweb service at http://cdaweb.gsfc.nasa.gov/. The authors would like to thank the AMPERE Science Center and NASA/GSFC for providing the data.

References

- Abel, G. A., Freeman, M. P., & Chisham, G. (2009). IMF clock angle control of multifractality in ionospheric velocity fluctuations. Geophysical Research Letters, 36(19). https://doi.org/10.1029/2009g1040336
- Abel, G. A., Freeman, M. P., Chisham, G., & Watkins, N. W. (2007). Investigating turbulent structure of ionospheric plasma velocity using the Halley SuperDARN radar. Nonlinear Processes in Geophysics, 14(6), 799-809. https://doi.org/10.5194/npg-14-799-2007
- Anderson, B. J., Angappan, R., Barik, A., Vines, S. K., Stanley, S., Bernasconi, P. N., & Barnes, R. J. (2021). Iridium Communications satellite constellation data for study of Earth's magnetic field. Geochemistry, Geophysics, Geosystems, 22(8), e2020GC009515. https://doi. org/10.1029/2020gc009515
- Anderson, B. J., Korth, H., Waters, C. L., Green, D. L., Merkin, V. G., Barnes, R. J., & Dyrud, L. P. (2014). Development of large-scale Birkeland currents determined from the Active Magnetosphere and Planetary Electrodynamics Response Experiment. Geophysical Research Letters, 41(9), 3017-3025. https://doi.org/10.1002/2014GL059941
- Anderson, B. J., Korth, H., Waters, C. L., Green, D. L., & Stauning, P. (2008). Statistical Birkeland current distributions from magnetic field observations by the Iridium constellation. Annales Geophysicae, 26(3), 671-687. https://doi.org/10.5194/angeo-26-671-2008
- Anderson, B. J., Takahashi, K., & Toth, B. A. (2000). Sensing global Birkeland currents with Iridium engineering magnetometer data. Geophysical Research Letters, 27(24), 4045-4048. https://doi.org/10.1029/2000GL00009410.1029/2000GL000094
- Barbosa, C. S., Caraballo, R., Alves, L. R., Hartmann, G. A., Beggan, C. D., Viljanen, A., et al. (2017). The Tsallis statistical distribution applied to geomagnetically induced currents. Space Weather, 15(9), 1094-1101. https://doi.org/10.1002/2017sw001631

Birkeland, K. (1908). The Norwegian Aurora Polaris Expedition 1902-1903. (Vol. 1). Christiania, Norway: H. Aschelhoug & Co.

- Burrell, A. G., Chisham, G., Milan, S. E., Kilcommons, L., Chen, Y.-J., Thomas, E. G., & Anderson, B. (2020). AMPERE polar cap boundaries. Annales Geophysicae, 38(2), 481-490. https://doi.org/10.5194/angeo-38-481-2020
- Chapman, S., & Ferraro, V. C. A. (1933). A new theory of magnetic storms. Terrestrial Magnetism and Atmospheric Electricity, 38(2), 79-96. https://doi.org/10.1029/TE038i002p00079
- Chisham, G., & Freeman, M. P. (2010). On the non-Gaussian nature of ionospheric vorticity. Geophysical Research Letters, 37(12). https://doi. org/10.1029/2010g1043714
- Chisham, G., & Freeman, M. P. (2021). A statistical model of vorticity in the polar ionosphere and implications for extreme values. Journal of Geophysical Research: Space Physics, 126(11), e2021JA029307. https://doi.org/10.1029/2021ja029307

Acknowledgments

J.C. Coxon was supported by Science and Technology Facilities Council (STFC) Ernest Rutherford Fellowship ST/V004883/1 and Consolidated Grant ST/R000719/1. G. Chisham and M.P. Freeman were supported by the British Antarctic Survey (BAS) Polar Science for Planet Earth Programme, as funded by the UK Natural Environment Research Council (NERC), as part of United Kingdom Research and Innovation (UKRI).

21699402, 2022, 2, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2021JA029801 by Test, Wiley Online Library on [21/06/2024]. See the Terms

and Conditions

Wiley Online Library for

. rules

of use; OA

articles are

governed by the applicable Creative Commons Licens



- Chisham, G., Freeman, M. P., Abel, G. A., Bristow, W. A., Marchaudon, A., Ruohoniemi, J. M., & Sofko, G. J. (2009). Spatial distribution of average vorticity in the high-latitude ionosphere and its variation with interplanetary magnetic field direction and season. *Journal of Geophysical Research*, 114(A9). https://doi.org/10.1029/2009JA014263
- Chisham, G., Lester, M., Milan, S. E., Freeman, M. P., Bristow, W. A., Grocott, A., et al. (2007). A decade of the Super Dual Auroral Radar Network (SuperDARN): Scientific achievements, new techniques and future directions. *Surveys in Geophysics*, 28(1), 33–109. https://doi. org/10.1007/s10712-007-9017-8
- Christiansen, F., Papitashvili, V. O., & Neubert, T. (2002). Seasonal variations of high-latitude field-aligned currents inferred from Ørsted and Magsat observations. Journal of Geophysical Research, 107(A2), 5–1. https://doi.org/10.1029/2001JA900104
- Clausen, L. B. N., Baker, J. B. H., Ruohoniemi, J. M., Milan, S. E., & Anderson, B. J. (2012). Dynamics of the region 1 Birkeland current oval derived from the Active Magnetosphere and Planetary Electrodynamics Response Experiment (AMPERE). Journal of Geophysical Research, 117, A06233. https://doi.org/10.1029/2012ja017666
- Clausen, L. B. N., Baker, J. B. H., Ruohoniemi, J. M., Milan, S. E., Coxon, J. C., Wing, S., et al. (2013). Temporal and spatial dynamics of the regions 1 and 2 Birkeland currents during substorms. *Journal of Geophysical Research: Space Physics*, 118(6), 3007–3016. https://doi. org/10.1002/jgra.50288
- Clausen, L. B. N., Milan, S. E., Baker, J. B. H., Ruohoniemi, J. M., Glassmeier, K.-H., Coxon, J. C., & Anderson, B. J. (2013). On the influence of open magnetic flux on substorm intensity: Ground- and space-based observations. *Journal of Geophysical Research: Space Physics*, 118(6), 2958–2969. https://doi.org/10.1002/jgra.50308
- Cowley, S. W. H. (2000). Magnetosphere-ionosphere interactions: A tutorial review. In Ohtani, S. -I., Fujii, R., Hesse, M., & Lysak, R. L. (Eds.), *Magnetospheric current systems* (Vol. 118, pp. 91–106). Washington, DC, USA: American Geophysical Union. https://doi.org/10.1029/gm118p0091
- Cowley, S. W. H., & Lockwood, M. (1992). Excitation and decay of solar wind-driven flows in the magnetosphere-ionosphere system. Annales Geophysicae, 10, 103–115.
- Coxon, J. C., Milan, S. E., & Anderson, B. J. (2018). A review of Birkeland current research using AMPERE. In Keiling, A., Marghitu, O., & Wheatland, M. (Eds.), *Electric currents in geospace and beyond* (pp. 257–278). American Geophysical Union (AGU). https://doi. org/10.1002/9781119324522.ch16
- Coxon, J. C., Milan, S. E., Carter, J. A., Clausen, L. B. N., Anderson, B. J., & Korth, H. (2016). Seasonal and diurnal variations in AMPERE observations of the Birkeland currents compared to modeled results. *Journal of Geophysical Research: Space Physics*, 121(5), 4027–4040. https://doi.org/10.1002/2015JA022050
- Coxon, J. C., Milan, S. E., Clausen, L. B. N., Anderson, B. J., & Korth, H. (2014a). The magnitudes of the regions 1 and 2 Birkeland currents observed by AMPERE and their role in solar wind-magnetosphere-ionosphere coupling. *Journal of Geophysical Research: Space Physics*, 119(12), 9804–9815. https://doi.org/10.1002/2014JA020138
- Coxon, J. C., Milan, S. E., Clausen, L. B. N., Anderson, B. J., & Korth, H. (2014b). A superposed epoch analysis of the regions 1 and 2 Birkeland currents observed by AMPERE during substorms. *Journal of Geophysical Research: Space Physics*, 119(12), 9834–9846. https:// doi.org/10.1002/2014JA020500
- Coxon, J. C., Rae, I. J., Forsyth, C., Jackman, C. M., Fear, R. C., & Anderson, B. J. (2017). Birkeland currents during substorms: Statistical evidence for a Region 2 intensification after onset and a local reduction in density before onset. *Journal of Geophysical Research: Space Physics*, 122, 6455–6468. https://doi.org/10.1002/2017JA023967
- Eastwood, J. P., Hapgood, M. A., Biffis, E., Benedetti, D., Bisi, M. M., Green, L., et al. (2018). Quantifying the economic value of space weather forecasting for power grids: An exploratory study. *Space Weather*, 16(12), 2052–2067. https://doi.org/10.1029/2018sw002003
- Freeman, M. P., Forsyth, C., & Rae, I. J. (2019). The influence of substorms on extreme rates of change of the surface horizontal magnetic field in the United Kingdom. Space Weather, 17(6), 827–844. https://doi.org/10.1029/2018sw002148
- Freeman, M. P., Southwood, D. J., Lester, M., & Waldock, J. A. (1990). Measurement of field-aligned currents by the SABRE coherent scatter radar. In Russell, C. T., Priest, E. R., & Lee, L. C. (Eds.), *Physics of magnetic flux ropes* (Vol. 58, pp. 575–580). American Geophysical Union. https://doi.org/10.1029/gm058p0575
- Iijima, T., & Potemra, T. A. (1976a). The amplitude distribution of field-aligned currents at northern high latitudes observed by Triad. Journal of Geophysical Research, 81, 2165–2174. https://doi.org/10.1029/ja081i013p02165
- Iijima, T., & Potemra, T. A. (1976b). Field-aligned currents in the dayside cusp observed by Triad. Journal of Geophysical Research, 81, 5971–5979. https://doi.org/10.1029/ja081i034p05971
- Iijima, T., & Potemra, T. A. (1978). Large-scale characteristics of field-aligned currents associated with substorms. Journal of Geophysical Research, 83(A2), 599–615. https://doi.org/10.1029/ja083ia02p00599
- Juusola, L., Milan, S. E., Lester, M., Grocott, A., & Imber, S. M. (2014). Interplanetary magnetic field control of the ionospheric field-aligned current and convection distributions. *Journal of Geophysical Research: Space Physics*, 119(4), 3130–3149. https://doi.org/10.1002/2013JA019455
- Knipp, D., Kilcommons, L., Hairston, M., & Coley, W. R. (2021). Hemispheric asymmetries in Poynting flux derived from DMSP spacecraft. Geophysical Research Letters, 48(17), e2021GL094781. https://doi.org/10.1029/2021gl094781
- Knipp, D. J., Matsuo, T., Kilcommons, L., Richmond, A., Anderson, B., Korth, H., et al. (2014). Comparison of magnetic perturbation data from LEO satellite constellations: Statistics of DMSP and AMPERE. Space Weather, 12(1), 2–23. https://doi.org/10.1002/2013SW000987
- Liu, J., Lyons, L. R., Wang, C.-P., Ma, Y., Strangeway, R. J., Zhang, Y., & Khurana, K. (2021). Embedded Regions 1 and 2 field-aligned currents: Newly recognized from low-altitude spacecraft observations. *Journal of Geophysical Research: Space Physics*, 126(6), e2021JA029207. https://doi.org/10.1029/2021ja029207
- Lockwood, M., Cowley, S. W. H., & Freeman, M. P. (1990). The excitation of plasma convection in the high-latitude ionosphere. *Journal of Geophysical Research*, 95(A6), 7961–7972. https://doi.org/10.1029/ja095ia06p07961
- Lockwood, M., & Freeman, M. P. (1989). Recent ionospheric observations relating to solar-wind-magnetosphere coupling. *Philosophical Transactions of the Royal Society of London Series A: Mathematical and Physical Sciences*, 328(1598), 93–105. https://doi.org/10.1098/ rsta.1989.0026
- Milan, S. (2019). AMPERE R1/R2 FAC Radii (Vol. 12). Retrieved From https://leicester.figshare.com/articles/dataset/ AMPERE_R1_R2_FAC_radii/11294861
- Milan, S. E., Carter, J. A., Korth, H., & Anderson, B. J. (2015). Principal component analysis of Birkeland currents determined by the Active Magnetosphere and Planetary Electrodynamics Response Experiment. *Journal of Geophysical Research: Space Physics*, 120, 10415–10424. https://doi.org/10.1002/2015JA021680
- Milan, S. E., Provan, G., & Hubert, B. (2007). Magnetic flux transport in the Dungey cycle: A survey of dayside and nightside reconnection rates. Journal of Geophysical Research, 112(A1), A01209. https://doi.org/10.1029/2006JA011642



Ngwira, C. M., & Pulkkinen, A. A. (2019). An introduction to geomagnetically induced currents. In *Geomagnetically induced currents from the* sun to the power grid (pp. 1–13). American Geophysical Union (AGU). https://doi.org/10.1002/9781119434412.ch1

Pakhotin, I. P., Mann, I. R., Xie, K., Burchill, J. K., & Knudsen, D. J. (2021). Northern preference for terrestrial electromagnetic energy input from space weather. *Nature Communications*, 12(1), 199. https://doi.org/10.1038/s41467-020-20450-310.1038/s41467-020-20450-3

- Pirjola, R. (2002). Review on the calculation of surface electric and magnetic fields and of geomagnetically induced currents in ground-based technological systems. Surveys in Geophysics, 23(1), 71–90. https://doi.org/10.1023/a:1014816009303
- Rogers, N. C., Wild, J. A., Eastoe, E. F., Gjerloev, J. W., & Thomson, A. W. P. (2020). A global climatological model of extreme geomagnetic field fluctuations. *Journal of Space Weather and Space Climate*, 10, 5. https://doi.org/10.1051/swsc/2020008
- Siscoe, G. L., & Huang, T. S. (1985). Polar cap inflation and deflation. Journal of Geophysical Research, 90(A1), 543-547. https://doi. org/10.1029/JA090iA01p00543
- Smith, A. W., Freeman, M. P., Rae, I. J., & Forsyth, C. (2019). The influence of sudden commencements on the rate of change of the surface horizontal magnetic field in the United Kingdom. Space Weather, 17(11), 1605–1617. https://doi.org/10.1029/2019sw002281
- Sofko, G. J., Greenwald, R., & Bristow, W. (1995). Direct determination of large-scale magnetospheric field-aligned currents with SuperDARN. Geophysical Research Letters, 22(15), 2041–2044. https://doi.org/10.1029/95GL01317
- Thomson, A. W., McKay, A. J., & Viljanen, A. (2009). A review of progress in modelling of induced geoelectric and geomagnetic fields with special regard to induced currents. Acta Geophysica, 57(1), 209–219. https://doi.org/10.2478/s11600-008-0061-7
- Thomson, A. W. P., Dawson, E. B., & Reay, S. J. (2011). Quantifying extreme behavior in geomagnetic activity. Space Weather, 9(10). https://doi.org/10.1029/2011sw000696
- Waters, C. L., Anderson, B. J., Green, D. L., Korth, H., Barnes, R. J., & Vanhamäki, H. (2020). Science data products for ampere. In Dunlop, M. W., & Lühr, H. (Eds.), *Ionospheric multi-spacecraft analysis tools: Approaches for deriving ionospheric parameters* (pp. 141–165). Springer International Publishing. https://doi.org/10.1007/978-3-030-26732-2_7
- Waters, C. L., Anderson, B. J., & Liou, K. (2001). Estimation of global field-aligned currents using the Iridium System magnetometer data. Geophysical Research Letters, 28(11), 2165–2168. https://doi.org/10.1029/2000GL012725
- Weimer, D. R. (2001). Maps of ionospheric field-aligned currents as a function of the interplanetary magnetic field derived from Dynamics Explorer 2 data. *Journal of Geophysical Research*, *106*(A7), 12889–12902. https://doi.org/10.1029/2000JA000295
- Workayehu, A. B., Vanhamäki, H., & Aikio, A. T. (2019). Field-aligned and horizontal currents in the northern and southern hemispheres from the Swarm satellite. *Journal of Geophysical Research: Space Physics*, 124(8), 7231–7246. https://doi.org/10.1029/2019ja026835
- Workayehu, A. B., Vanhamäki, H., & Aikio, A. T. (2020). Seasonal effect on hemispheric asymmetry in ionospheric horizontal and field-aligned currents. Journal of Geophysical Research: Space Physics, 125(10), e2020JA028051. https://doi.org/10.1029/2020ja028051
- Xiong, C., Stolle, C., Alken, P., & Rauberg, J. (2020). Relationship between large-scale ionospheric field-aligned currents and electron/ion precipitations: DMSP observations. *Earth Planets and Space*, 72(1), 147. https://doi.org/10.1186/s40623-020-01286-z